 Vc/22-Mar-04
Week 13  starting 22 March 04

SM18 Operation Meeting (Tuesdays)

 Bench status  and tests for this week : 

	Bench
	Name
	 
	Temp
	-
	-

	TBC1
	MBAL2066
	NEW Arrival
	300 K
	 
	 

	TBC2
	SSS013
	ICS 2 Connect Magnet (ICS)
	300 K
	 
	 

	TBD1
	MBAR3050
	Next in line for cooldown?
	300 K
	 
	 

	TBD2
	MBAR2005
	 
	Cooling
	 
	 

	TBE1
	MBBR2048
	NEW Arrival
	Warming
	 
	 

	TBE2
	MBBR3003
	Next in line for cooldown?
	300 K
	 
	 

	TBF1
	MBAR2011
	 
	Cooling
	 
	 

	TBF2
	MBBL3078
	Finishing? Next 2016  Shafts
	 
	 
	 


Test Coordinator this week:  Vittorio Remondino; MTM support on call Giorgio:  (16-3602).

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
	Date
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday 16_M


	TBD2: T coils crate 2 slot 8 : all channels showing negative values. 

Fault temperature in sector 12, coil e (shaft 30).
	MTM
	LF
	Cards in LF crate inverted. Guy solved the problem. 

Coil e of sector 12 is broken..

	Tuesday 16_M


	TBD2: During IAP, Qloc card problem.
	MTM
	
	Giorgio and Guy changed the card.

	Tuesday 16_M


	TBF1: Final warm up of 3056

Lost of the cryo OK during the Min Energy.
	ACR
	
	

	Thursday 18_M


	TBD2: Problem to warm up the 3010.

1) Heaters did not work

2) Pollution of the Helium buffer (Liquefactor). CWU has to be regenerated


	ACR


	
	Action taken by J.Paul Lamboy, A.Tovar.

~7 hours lost.



	Thursday 18_M


	TBF1, F2: ICS received wrong information about the status of the magnets 2069 (TBF2) and 3056 (TBF1).

2069 has to be in stand by (and not stripped), 3056 is sent for stripping (and not in stand by).
	MTM
	IAP
	Stephane clarifies the situation with B.Poulnais. Correct information was written in the paper traveller (Daniel). Investigation by J.Axensalva to see if the mistake occurred in the TTS.

	Thursday 18_N

Friday _19 _M


	TBD1: During preparation of HV test@ cold cryo-team received a quench signal.

Same problem in TBC1.

Origin: Lost of communication in the PLC cryo.
	ACR


	
	Tests HV and IAP done with the supervision of the equip.support.

Problem solved by the cryo team.

	Friday _19 _M


	TBF1: (2011) During IAP oscillation of the temperature at warm with an amplitude of 1 K.
	MTM
	IAP
	Guy called. He  considered as acceptable and gave the OK.

	Friday _19 _A


	Problem to upload files in MTF.

Error message : File name did not accept space.


	MTM
	
	Space in the name cancelled, then OK.

Message : No space is allowed in the file names for MTF.

	Friday _19 _N


	TBD1, TBD2: Display box “high current” did not glow when the converter is on.


	MTM
	
	

	Friday _19 _N


	TBE: Discrepancy between the reading of the position of the rotary switch in the Cryo TV screen (E2 position) and the real position on the bench  (CC position). The same for the F bench.


	ACR
	
	

	Friday _19 _N


	TBD1: HV final cold. Quench signal received by Cryo Team. Communication problems which triggers quench signals. 


	ACR
	
	Solved 4 hours later by ACR expert?

	Saturday _20 _A


	TBC1: Water interlock on Power Converter.

The valve was not opened.


	MTM
	
	Guy came and opened the valve.

In that case, please check if the valve is open before calling the support (Guy).

	Saturday _20 _N


	TBC1: Bad rotation of the shafts. 


	MTM
	
	Guy called. He found that in the TRU 4, the coupler between the encoder and the potentiometer was broken. He found a new coupler and changed it. For the future a procedure has to be implemented with P.Legrand to help the equip.support for this type of intervention.

	Sunday_21_M


	TBE1: AQA error  in the LF quench heater discharge during the quench at 1.5 kA.

AQA showed  2 quench heater not discharged.
	MTM
	AQA
	Guy called. Test repeated. AQA showed OK but quench table was missing. Guy and Giorgio investigate the origin of the problem monday.

	Sunday_21_M


	TBC1: Lost of the cryo OK during the LHC.


	ACR
	
	

	Sunday_21_A


	TBC1:Power application not running in TEMA as well in local station.

Error was : Vi missing.


	AB-CO
	
	Guy called Adrian. Adrian erased some files then OK.

Origin : disk full?








